Author's response to reviews

Title: Locally advanced breast cancer made amenable to radical surgery after a combination of systemic therapy and Mohs paste: A case series

Authors:

Tomoya Tsukada (tkd_tmy@nifty.com)
Tatsuo Nakano (nakano@asanogawa-gh.or.jp)
Miki Matoba (mikimtb1231@gmail.com)
Daisuke Matsui (bokyahinnn0914@gmail.com)
Shozo Sasaki (syozossjp@yahoo.co.jp)

Version: 3 Date: 24 September 2012

Author's response to reviews: see over
September 24, 2012

Professor Michael Kidd AM
Editor-in-Chief
Journal of medical case reports

Dear Prof. Kidd:

Re: Manuscript reference no. MS 1991311853702546

Please find attached a revised version of our manuscript “Locally advanced breast cancer made amenable to radical surgery after a combination of systemic therapy and Mohs paste: A case series”, which we would like to resubmit for publication as a case report series in Journal of medical case reports.

Your comments were very insightful and enabled us to greatly improve the quality of our manuscript.

Revisions in the text are shown using yellow highlights for additions. In accordance with editor’s suggestion, we added a short statement in Figure legends part. We also added the allowheads on Figures 1b and 2b. We hope that the revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript suitable for publication in Journal of medical case reports.

Thank you for your consideration. We look forward to hearing from you at your earliest convenience.

Sincerely,
Tomoya Tsukada M.D., Ph.D.
Department of Gastroenterological Surgery, Division of Cancer Medicine
Graduate School of Medical Science, Kanazawa University
13-1 Takara-machi, Kanazawa
Ishikawa 920-8641
Japan

Phone: 81-76-265-2362
Fax: 81-76-234-4260
E-mail: tkd_tmy@nifty.com